

Citation: Haque F, Lampe FC, Hajat S, Stavrianaki K, Hasan SMT, Faruque ASG, et al. (2024) Impacts of climate change on diarrhoeal disease hospitalisations: How does the global warming targets of 1.5–2°C affect Dhaka, Bangladesh? PLoS Negl Trop Dis 18(9): e0012139. https://doi.org/ 10.1371/journal.pntd.0012139

Editor: James Michael Fleckenstein, Washington University School of Medicine in Saint Louis: Washington University in St Louis School of Medicine, UNITED STATES OF AMERICA

Received: April 8, 2024

Accepted: August 20, 2024

Published: September 26, 2024

Copyright: © 2024 Haque et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Excel spreadsheet (<u>S1 Data</u>) containing, in separate sheets, the underlying numerical data and statistical analysis for Figs <u>1–5</u> is attached. According to institutional data policy of the icddr,b (International Centre for Diarrhoeal Disease Research, Bangladesh), only summary of patient data can be publicly displayed or can be made publicly accessible. To protect intellectual property rights of primary data, icddr,b

RESEARCH ARTICLE

Impacts of climate change on diarrhoeal disease hospitalisations: How does the global warming targets of 1.5–2°C affect Dhaka, Bangladesh?

Farhana Haque^{1,2*}, Fiona C. Lampe¹, Shakoor Hajat³, Katerina Stavrianaki⁴, S. M. Tafsir Hasan⁵, A. S. G. Faruque⁵, Tahmeed Ahmed⁵, Shamim Jubayer⁶, Ilan Kelman^{7,8}

 Institute for Global Health (IGH), University College London (UCL), London, United Kingdom, 2 UK Public Health Rapid Support Team (UK-PHRST), Department of Infectious Disease Epidemiology, London School of Hygiene and Tropical Medicine (LSHTM), London, United Kingdom, 3 Department of Public Health, Environments and Society, London School of Hygiene and Tropical Medicine (LSHTM), London, United Kingdom, 4 Department of Statistical Science, Department of Risk and Disaster Reduction, University College London (UCL), London, United Kingdom, 5 Nutrition and Clinical Services Division, icddr,b, Dhaka, Bangladesh, 6 National Heart Foundation Hospital and Research Institute (NHF&RI), Dhaka, Bangladesh,
7 Institute for Global Health (IGH) and Department of Risk and Disaster Reduction, University College London (UCL), London, United Kingdom, 8 University of Agder, Kristiansand, Norway

* farhana.haque@lshtm.ac.uk, farhana.haque@alumni.ucl.ac.uk

Abstract

Dhaka is one of the world's densely populated cities and faces significant public health challenges including high burden of diarrhoeal diseases. Climate change is intensifying existing environmental problems including urban heat island effect and poor water quality. While numerous epidemiological studies have linked meteorological factors to diarrhoeal diseases in Bangladesh, assessment of the impacts of future climate change on diarrhoeal diseases is scarce. We provide the assessment of climate change impacts on diarrhoeal disease in Dhaka and project future health risks under climate change scenarios. About 3 million acute diarrhoea cases presenting to the Dhaka Hospital of the International Centre for Diarrhoeal Disease Research Bangladesh (icddr,b) during 1981–2010 were linked to daily temperature, rainfall and humidity and association investigated using time series adapted negative binomial regression models employing constrained distributed lag linear models. The findings were applied to climate projections to estimate future risks of diarrhoea under various global warming scenarios. There was a significantly raised risk of diarrhoea hospitalisation in all ages with daily mean temperature (RR: 3.4, 95% CI: 3.0-3.7) after controlling for the confounding effects of heavy rainfall, humidity, autocorrelations, day of the week effect, long-term time, and seasonal trends. Using the incidence rate ratio (IRR) of 1.034, temperature increases based on the global warming targets of 1.5-2°C could result in an increase of diarrhoea hospitalisations by 4.5-7.4% in all age groups by the 2100s. These effects were more pronounced among <5 children where the predicted temperature increases could raise diarrhoea hospitalisation by 5.7% - 9.4%. Diarrhoea hospitalisation will increase significantly in Dhaka even if the global warming targets adopted by the Paris Agreement is

cannot make primary data publicly available. However, upon request, Institutional Data Access Committee of icddr,b can provide access to primary data to any individual, upon reviewing the nature and potential use of the data. Requests for data can be forwarded to: Ms. Armana Ahmed, Head, Research Administration, icddr,b, Dhaka, Bangladesh, Email: aahmed@icddrb.org, info@icddrb.org, Phone: +88 02 9827001-10 (ext. 3200).

Funding: This study was completed as a part of a PhD thesis by FH. The authors would like to thank the Commonwealth Scholarship Commission for funding the PhD study. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

reached. This underscores the importance of preparing the city for management and prevention of diarrhoeal diseases.

Author summary

Dhaka, a resource-constrained, crowded city in Southeast Asia, struggles with serious public health issues, especially a high number of diarrhoeal diseases. Climate change is worsening Dhaka's environmental problems by making Dhaka hotter and deteriorating water quality. While some studies have already looked at how weather affects diarrhoea in Bangladesh, few have examined the future impact of climate change. This study explored how rising temperatures might affect diarrhoea in Dhaka in the future. Using advanced statistical methods, the researchers analysed information from nearly 3 million diarrhoea patients treated at a major hospital in Dhaka from 1981 to 2010. The study found that higher daily temperatures significantly increased the risk of diarrhoea for all age groups, even after considering other factors like heavy rain, humidity, and seasonal patterns. Based on future temperature predictions, even if global warming is limited to 1.5 to 2°C, as targeted by international agreements, diarrhoea cases could rise by 4.5% to 7.4% by the end of the century. The increase could be even higher for children under 5 years old, with hospital admissions going up by 5.7% to 9.4%. This highlights the need for Dhaka city to better prepare for managing and preventing diarrhoeal diseases in the future.

Introduction

The global mean surface temperature is projected to rise by 2.6°C–4.8°C under the high emissions scenarios in 2100 even if all countries reduce their greenhouse gas emissions as pledged in the Paris Agreement [1]. Such temperature increase poses significant threat to the human society as global warming can adversely affect human health and wellbeing, particularly in the low- and middle-income countries (LMICs) that have little economic capacity to take adequate mitigation and adaptation measures [2–4]. Climate change is projected to exert significant impact on hydrological systems, water availability and water quality. As a result, waterborne diseases, particularly diarrhoeal diseases are expected to be adversely affected by the impacts of climate change [5,6]. Diarrhoeal diseases, with an estimated annual global incidence of 2.39 billion episodes and 1.3 million deaths in all ages according to the global burden of disease study in 2015, remain one of the leading causes of morbidity and mortality worldwide [7]. Given the high baseline rates, even a small increase in risk stemming from climate change can significantly increase public health burden in the LMICs.

Despite the concerns that climate change can cause excess diarrhoea, only a limited number of studies have attempted to quantify the effects of climate change on diarrhoea across the globe with variable results [8–10]. Using projections from one climate model and empirical data from two studies conducted in Fiji [11] and Peru [12], one study inferred that global warming of 1°C by the 2030s will lead to an increase in diarrhoea by 0–10% [8]. Another study projected increases of 8–11% in relative risks of diarrhoea by 2010–2039 using data from five empirical studies across the globe and 19-member climate model ensemble [9]. On the other hand, Moors et al. (2013) calculated increases of 0–21% in diarrhoea by the 2040s, using the latest regional climate projections for northern India [10]. A recent study projected over a 10% rise in heat-related diarrhoeal diseases in the Gaza Strip, Palestine under the 2°C global warming scenario [13].

Bangladesh is a low- and middle-income country (LMIC) in Southeast Asia with a lowlying topography, poor infrastructure and high population density [14]. Bangladesh has been classified in the Sixth Assessment Report of the Intergovernmental Panel on Climate Change (IPCC) as a country that is highly vulnerable to the adverse impacts of climate change [1]. As temperatures continue to rise and rainfall becomes more erratic, water quality, water stress, food safety and food security may deteriorate in Bangladesh [14-16]. Though Bangladesh has made strong progress towards ensuring better quality drinking water-the switching to improved water infrastructure-much of what is tapped remains contaminated with dangerous microbes, heavy metals, or salt. Moreover, infrastructure alone may not ensure quality of drinking water supplies at all times [17]. Diarrhoeal disease is endemic in Bangladesh. While the exact prevalence estimate of diarrhoea is lacking in Bangladesh, diarrhoea is a major cause of morbidity in the country [7]. According to the Bangladesh Demographic and Health Surveys (BDHS), the reported prevalence of diarrhoea in the previous two weeks among underfive children was 5.7% and the prevalence of diarrhoea with blood was 1% in 2014. Despite meeting the water MDG target successfully in 2010, outbreaks and hyper-endemicity of diarrhoeal diseases continue to plague the developing nations including Bangladesh [18]. With more than an estimated 76 million diarrhoeal disease episodes in all age groups in Bangladesh during 2015 [7], the potential impact of long term variability in the climate on the incidence of diarrhoeal disease in the future could be concerning for Bangladesh.

This study aims to assess the impacts of meteorological factors on acute diarrhoeal disease in Dhaka, Bangladesh using daily time-series datasets and quantify the effects of temperature increase under future climate change scenarios.

Materials and methods

Ethics statement

This study was granted approval by the Research Review Committee (RRC) and Ethical Review Committee (ERC) of the International Centre for Diarrhoeal Diseases Research, Bangladesh (icddr,b) (Protocol # PR-19097). The study used secondary data and did not involve primary data collection from human participants. The study was also approved by the UCL Research Ethics Committee (UCL REC).

Study setting

Dhaka, located at 23°42′N 90°22′E on the eastern banks of the Buriganga River, is the capital and largest city of Bangladesh. Tropical vegetation and moist soils characterize the land, which is flat and close to sea level. Dhaka, a low-lying, predominantly coastal city surrounded by five major rivers and 50 water channels, is at risk of flooding from rising sea levels [19]. Several dykes and embankments and interim boundaries created to protect Dhaka from flooding together with inadequate drainage and water pumping systems frequently result in water log-ging of the protected city area during the monsoons [20]. With a density of 23,234 people per square kilometres within a total area of 300 square kilometres of the city area, Dhaka city faces a number of environmental and societal challenges including urban heat island effect, insufficient infrastructure, poor water quality, inadequate sanitation, inadequate waste management, poor hygiene brought about by poverty [21,22].

Health data

Data on the daily number of acute diarrhoea cases (defined as patients passing \geq 3 loose stools per 24h due to any cause) presenting to the Dhaka Hospital of the International Centre for

Diarrhoeal Diseases Research, Bangladesh (icddr,b) between 1981–2010 were obtained on 7 October 2020. The Dhaka hospital is the largest diarrhoeal disease hospital that predominantly served an urban population of approximately 14.7 million and provided free treatment to more than 140,000 patients with diarrhoea in 2010 [23]. Given that reliable records of the total number of patients with acute diarrhoea admitted per day or their disease onset dates were not available for the study period (1981–2010), information from the Diarrhoeal Disease Surveillance System (DDSS) was obtained instead to estimate the total number of patients hospitalised with diarrhoea per day. The DDSS was established in 1979 to record the information of a sample of acute diarrhoea patients who sought care from the icddr,b Dhaka Hospital [24]. During 1981–1995, the DDSS enrolled every 25th patient and during 1996–2010, the DDSS enrolled every 50th patient who were admitted in the icddr,b Dhaka Hospital into the surveillance system. The Dhaka Hospital began keeping electronic records of all patients seeking care since 2010 [25]. We did not access any information that could identify individual participants during or after data collection. We did not collect any data beyond the study period. Additional information on health data is provided in S1 File.

Meteorological data

Daily climate parameters including mean, maximum and minimum temperature, cumulative rainfall, and relative humidity between 1 January 1980 and 31 December 2010 for Dhaka City were collected from the Bangladesh Meteorological Department (BMD). The BMD maintains 3-hourly records of temperature data from three validated weather stations across Dhaka. All data are publicly accessible at https://bmd.gov.bd/.

While climate projections for Dhaka city are limited, a study projected an increase in maximum temperature by 1.3-4.3 °C in Bangladesh for different representative concentration pathway (RCP) scenarios using the Coupled Model Intercomparison Project Phase 5 (CMIP5) global climate models (GCMs) models by the 2100 [26]. Various regional climate models (RCMs) using different IPCC greenhouse gas emission scenarios predicted a likely increase in the mean temperature by 1–3°C during 2030–2060 in Bangladesh [27]. Another study projected change of mean surface air temperature in different months by 0.5–2.1 °C and 0.9–3.5 °C for the year 2050 and 2060, respectively [28]. For the SRES A1B emissions scenario (representing a future world of very rapid economic growth, low population growth and rapid introduction of new and more efficient technology), the projected temperature increases over the entire country are in the region of 3 to 3.5°C by the 2100s according to the UK's Met Office Hadley Centre [29]. On the other hand, a study based on an ensemble of projections using the high resolution (10 km) MIT Regional Climate Model (MRCM) projected an increase of 1.2-1.3°C in daily maximum temperature by the 2050s for Bangladesh compared to a reference period of 1976-2005. Using statistical downscaling model, one study predicted an increase of 2.7°C and 2.1°C in average annual maximum and minimum temperature per century [30]. Using a method of combining different independent models known as multi-model ensemble (MME) that synthesised information from 40 CMIP5 GCMs, a recent study showed that mean annual maximum temperature over Bangladesh will increase by 0.61–1.75°C in the near future (2021–2060) and 0.91–3.85°C in the far future (2071–2100) [14]. Given that different climate projections suggest a variable increase in daily temperature across Bangladesh ranging between 0.91-4.3°C by the 2100s and because climate projections are prone to biases [9], we estimated the increased risk of diarrhoea within the projected temperature range for Bangladesh. In addition, we estimated the increase in relative risk of diarrhoea hospitalisation for the global warming targets agreed in the Paris Agreement since many countries have reaffirmed their commitment to hold the increase in global average temperature to 1.5-2°C above the pre-industrial levels.

Additional data

Additional socio-economic data for sensitivity analyses were collected as follows: yearly total urban population projections for Dhaka was collected from the UN population prospects; the annual gross domestic product (GDP) per capita in Bangladesh was obtained from the Nation-master website; literacy rate was obtained from the Bangladesh Bureau of Statistics and the 5-yearly data on access to safe drinking water and sanitation coverage were obtained from the Demographic and Health Surveillance (DHS) and the Joint Monitoring Programme (JMP). To compare how diarrhoea hospitalisation rates compared with national trends, we additionally collected annual diarrhoea hospitalisation and death data from the website of the Department of Health Services of the Ministry of Health and Family Welfare, Bangladesh (https://old.dghs.gov.bd/index.php/en/publications/health-bulletin/dghs-health-bulletin).

Analysis

Epidemiological analysis to model baseline impacts. Negative binomial time-series regression models were used to assess acute associations between daily diarrhoea hospitalisation and daily ambient temperature, relative humidity and heavy rainfall (defined as the rainfall \geq 95th percentile for the study period). A series of linear-spline functions of time were used to flexibly model underlying long-term trends and seasonal patterns in the health data unrelated to climate factors [25,31]. Given that overdispersion or extra-variations in the outcome counts was detected, seasonally adjusted negative binomial model was used to allow for overdispersion. Model residuals were lagged by one day to adjust for the significant residual autocorrelation (i.e., serial dependence) associated with the day before. The conceptual framework for considering factors as potential confounders, covariates and/or effect modifiers in the models is provided in S2 File.

Distributed lag linear models (DLLMs) were then employed to model the delayed effects of each exposure that allowed the relationship to be simultaneously modelled at different lags of exposure. A maximum lag of 28 days was used to capture delayed effects. These models suggested a linear risk associated with mean temperature at lag 0–1 days, humidity at lag 0–3 days and heavy rainfall at lag 0–8 days. The cumulative effects of temperature, humidity or rainfall across all significant lags (p-value <0.05) were therefore estimated using DLLMs. Although each variable was investigated separately, the final quantification was based on a model with all meteorological factors simultaneously entered in a constrained DLLM, which took the following form:

$$\log[E(Y_t)] = \beta_0 + \sum \beta_1 Temp_{t-1} + \sum \beta_2 Hum_{t-3} + \sum \beta_3 HeavyRain_{t-8} + \beta_4 LS(Time_t, 8 D.F.) + \sum \beta_5 dow + \gamma r_{t-1}$$
(1)

where $E(Y_t)$ is the expected diarrhoea hospitalisation count on day t; $Temp_{t-1}$ indicates the distributed lag of mean temperature at lags 0–1 days; and Hum_{t-3} and $HeavyRain_{t-8}$ indicate the distributed lag of relative humidity and heavy rainfall at lags 0–3 and 0–8 days, respectively. *LS* indicates a linear spline of time with 8 degrees of freedom per year; *dow* is the categorical day of the week with a reference day of Friday; and r_{t-1} is deviance residuals at 1 day lag obtained from the model without any autoregressive terms. The relative risk was estimated as incidence rate ratio (IRR) in the model and later converted into percentage.

Multiple sensitivity analyses were conducted by changing the amount of control for seasonality and long-term trends in the model (by using natural cubic splines and changing the number of knots in the spline-based approach). Further sensitivity analysis was conducted by including population of Dhaka into the model as an offset (S3 File). The analysis was also repeated using the actual number of diarrhoeal disease patients enrolled into the DDSS during the study period. In addition, two separate models using a generalized estimating equation (GEE) approach were generated to investigate the potential modification by improved WASH interventions including access to safe drinking water and sanitation coverage [32]. Access to safe drinking water is measured as the percentage of population in an area drinking water from an improved source that is accessible on premises, available when needed and free from faecal and priority chemical contamination. Improved water sources include piped water, boreholes or tubewells, protected dug wells, protected springs, rainwater, and packaged or delivered water. Sanitation coverage is defined as the percentage of households in an area with access to sanitation facilities based on the Joint Monitoring Program (JMP) service levels criteria developed by the United Nations. Diagnostic plots based on deviance residuals were generated for each model to check the model fit. The best model was chosen based on the lowest BIC value (S3 File). All data were analysed using Stata/SE 18.0 (StataCorp, USA). Additional descriptions of the scientific terms are provided in S4 File.

Estimating the impact of climate change on diarrhoea. The impact (I) of climate change on diarrhoea hospitalisation in the future was calculated by multiplying the projected increase in temperature (Δ T) with the percentage increase in relative risk (RR) of diarrhoea hospitalisations with each 1°C temperature increase using the formula below:

$$\mathbf{I} = (1 + \mathbf{R}\mathbf{R}) \mathbf{X} \,\Delta \mathbf{T} \tag{2}$$

where RR is the increase in risk percentage of diarrhoea per 1°C increase in temperature and ΔT is the projected increase in temperature by the 2100s.

Results

During the 30-year study period between 1 January 1981 and 31 December 2010, approximately 2,983,850 cases of all ages presented to the icddr,b Dhaka Hospital for acute diarrhoeal disease. Among these, 58% were males and 54% were children aged <5 years. Figs 1 and 2 show the daily and monthly distributions of diarrhoea hospitalisation and mean temperature (°C) in Dhaka averaged across years 1981–2010. Diarrhoea hospitalisation for under 5 children peaked in April whereas diarrhoea hospitalisation for all ages revealed a large peak in April and a smaller peak in September (Fig 2). As highlighted in Fig 2, under-5 diarrhoea hospitalisation during the months with high rainfall (June and August) was similar to the hospitalisation during the months with low rainfall (December, January and March).

According to Fig 3, hospitalisation rates in the Dhaka Hospital showed a declining trend with improvements in water supply, sanitation and socio-economic conditions. As evident from Figs 2 and 4, diarrhoea morbidity rates remain significantly high both in Dhaka and across Bangladesh during the study period. The numerical data used in all figures are included in S1 Data.

Fig 5 shows the seasonally adjusted relationship between temperature with relative risk of diarrhoea hospitalisation in all ages across multiple lags based on the DLLM model. There was a linearly increased risk of diarrhoea hospitalisation associated with temperature on the same day and a decreased risk at lag 1 day with negligible effects observed until lags of 28 days. The relationship with rainfall exhibited a broadly linear increased diarrhoea hospitalisation risk with heavy rainfall and the effect persisting for up to 8 days following exposure. We observed no increased risk of diarrhoea hospitalisation with low rainfall. In addition, relative humidity was inversely associated with diarrhoea hospitalisations in all ages and \leq 5 children in Dhaka were estimated to increase by 3.4% (95% CI: 3.0–3.7) and 3.9% (3.8–4.7) per 1°C increase in



Fig 1. Daily distribution of diarrhoea hospitalisation and ambient temperature averaged across years 1981–2010. https://doi.org/10.1371/journal.pntd.0012139.g001



Fig 2. Monthly distribution of daily diarrhoea hospitalisation in all ages and \leq 5 children and meteorological factors averaged across years 1981–2010.

https://doi.org/10.1371/journal.pntd.0012139.g002



Fig 3. Demographic, economic, improved water supply access, sanitation coverage and diarrhoea hospitalisation trends in Dhaka, Bangladesh 1981–2010.

https://doi.org/10.1371/journal.pntd.0012139.g003

daily mean temperature, respectively (<u>Table 1</u>). The results did not change significantly during the sensitivity analyses (<u>S3 File</u>).

As shown in Table 2, diarrhoea hospitalisation in all ages is predicted to increase by 5.1% and 7.4% under the 1.5°C and 2°C GWLs, respectively. Diarrhoea hospitalisation in children under 5 years of age may be increased by 5.9% (95% CI: 5.7–7.1) under the 1.5°C and by 7.8% (95% CI: 7.6–9.4) under the 2°C GWL. With a projected rise in Dhaka's temperature by 4.3°C by the 2100, diarrhoea hospitalisation could rise by 15.9% and 20.2% in all ages and \leq 5 children, respectively. Compared to all age groups, temperature increases under all scenarios showed higher impacts in children under 5 years of age. The impact of temperature increase on diarrhoea hospitalisation varied by gender with a slightly higher influence in males compared to females (5.1–6.8% Vs 5.8–7.0%) (Table 2). If the baseline relationship from 2010

m 1		D /			1. 1		• • •		• •			• • •	• •		11	1				D 7		4
Tab	10 I	Unrconton	10 10 0000	0 110	diaret	loog h	ocnita.	1001101	1 000001010	A 11711	h 000 111	11 11	ncroaco in to	mnoratura	raintal	0r h11	midity	7 1 10 1	halza	Rong	10doc	•
1 4 1	ю.	reitentag	2C IIICI CAS	сш	וומוח	юса п	iusijita	пзаноі	i assuciaic	uwii	н онс ц	нк н	потсаяе ни не	CHINCIALUIC	, i aiiiiaii	UI IIU	munu		/IIakas	Dalle	Iducs	44
			,												,				,			

Subgroup	Temperature (Lag 0–1 days)		Rainfall (Lag 0-8 days)	Humidity (Lag 0-3 days)			
	Percentage increase per 1°C rise in daily mean temperature (95% CI)	<i>p</i> -value	Percentage increase per 1 mm rise in daily cumulative rainfall beyond 35 mm (95% CI)	<i>p</i> -value	Percentage decrease per 1% rise in daily relative humidity	<i>p</i> -value	
All ages	3.4 (3.0-3.7)	0.0001	2.6 (1.1–3.7)	0.0001	0.2 (0.1–0.2)	0.0001	
\leq 5 years	3.9 (3.8-4.7)	0.0001	2.8 (2.2–5.3)	0.0020	0.2 (0.1–0.3)	0.0001	
Females	3.4 (2.3-4.6)	0.0010	2.5 (0.9–3.9)	0.0007	0.1 (0.1–0.2)	0.0020	
Males	3.5 (2.7–4.3)	0.0005	2.7 (1.0-4.1)	0.0009	0.1 (0.1–0.2)	0.0021	

https://doi.org/10.1371/journal.pntd.0012139.t001



Fig 4. National trends of diarrhoea hospitalisation and deaths in Bangladesh during 2000–2016. https://doi.org/10.1371/journal.pntd.0012139.g004

between meteorological factors and diarrhoea hospitalisation risk and all other factors including population remain constant in the future, an increase in mean temperature by 0.91°C, 1.5°C, 2°C or 4.3°C could result in 4,760, 7,840, 10,360 and 22,260 additional diarrhoea hospitalisations annually in Dhaka Hospital by the 2100, respectively. Applying the same baseline





https://doi.org/10.1371/journal.pntd.0012139.g005

Subgroup	Relative risk (%) of diarrhoea hospitalisation								
	With 1.5°C GWL* (95% CI)	With 2°C GWL (95% CI)	With 0.91°C predicted increase by 2100 (95% CI)	With 4.3°C predicted increase by 2100 (95% CI)					
All ages	5.1 (4.5-5.6)	6.8 (6.0-7.4)	3.1 (2.7–3.4)	14.6 (12.9–15.9)					
≤5 children	5.9 (5.7-7.1)	7.8 (7.6–9.4)	3.6 (3.5-4.3)	16.8 (16.3–20.2)					
Females	5.1 (3.5-6.9)	6.8 (4.6-9.2)	3.1 (2.1–4.2)	14.6 (9.9–19.8)					
Males	5.3 (4.1-6.5)	7.0 (5.4–8.6)	3.2 (2.5–3.9)	15.1 (11.6–18.5)					

Table 2. Impacts of projected temperature increase on diarrhoea hospitalisation risk in Dhaka Bangladesh under various climate change scenarios.

*GWL: Global Warming Level targeted in the Paris Agreement

https://doi.org/10.1371/journal.pntd.0012139.t002

risk as Dhaka, a temperature increase of 4.3 °C by the 2100 could result in 385,955 additional hospitalisations across Bangladesh annually.

Discussion

While numerous studies have explored the effects of climate variability on diarrhoea worldwide [33], only a small number of studies have attempted to estimate the impacts of future climate change on diarrhoea [9,10,13]. Using empirical data from icddr,b Dhaka Hospital, our findings suggest significant baseline burden of acute diarrhoeal disease in Dhaka associated with ambient temperature after adjusting for the confounding effects of humidity, heavy rainfall, seasonal patterns, long-term trend, and day-of-the-week. This risk may aggravate further as climate continues to change. Our findings reveal that temperature-related diarrhoeal disease could increase by 7.4% in people of all ages and 9.4% in children under 5 years of age under the 2°C global warming level compared to the reference period of 1981–2010. This increase would be limited to 5.6% in people of all ages and 7.1% in children under 5 years of age if the 1.5°C global warming target could be reached. However, diarrhoeal disease hospitalisation could rise by up to 15.9% in all ages and 20.2% in children under 5 years if Dhaka's temperature increases by 4.3°C by the 2100 as projected by various climate models [26].

With notable improvements in socio-economic condition and healthcare provision and advances in water, sanitation and hygiene infrastructure in Bangladesh, diarrhoea mortality in children has declined significantly from 15.1 per 1000 in 1980 to 6.0 per 1000 live births in 2015 [34,35]. The rate of diarrhoea hospitalisation in Dhaka Hospital also suggests a declining trend. Despite success in reduction of diarrhoea mortality, outbreaks and hyperendemicity continue to plague the nation and diarrhoea remains a major cause of morbidity in Bangladesh for people in all age groups including children under 5 years of age [18,36].

The baseline risk of diarrhoea hospitalisation with temperature in Dhaka observed in this study was smaller compared to many studies [33]. Given the high baseline rates, the temperature-attributable numbers of diarrhoea hospitalisation were still substantially large. In this study, the observed effects of temperature on diarrhoea hospitalisation were mostly immediate with negligible effects observed for up to 28 lag days. Hot weather can promote environmental expansion of diarrhoeal pathogens, raise microbial load of drinking water, enhance food spoilage and increase consumption of contaminated food and water [5,37]. All these factors can contribute to a higher incidence of diarrhoea and the effect may be immediate since the incubation period may be as low as 2 hours for some diarrhoeal pathogens [38]. Severely affected diarrhoea patients often seek hospital care within 4–12 hours of symptom onset which may help explain the observed immediate effects of temperature on diarrhoea hospitalisation in this study [39].

In this study, diarrhoea hospitalisation decreased with mean temperature at the lag of 1-day following an initial increase in diarrhoea hospitalisation on the same day. In environmental epidemiological studies, short-term increase in risk of diarrhoea followed by a protective effect at a longer lag is known as 'harvesting' or 'short-term displacement' effect [31]. Prior studies eliciting the impact of climate parameters on diarrhoea rarely reported the occurrence of any harvesting effect. Aik et al (2019) examining climate variability on diarrhoea incidence in Singapore using time series analysis reported evidence of harvesting effect of diarrhoeal disease due to relative humidity six weeks later [40]. Another study investigating the influence of climate on diarrhoea in Vietnam reported evidence of harvesting effect due to rainfall but did not include the protective effect of rainfall into the final model [41]. Reductions in at-risk population following diarrhoea episodes likely contributed to the decreased diarrhoea hospitalisations observed the following day. Furthermore, family members and allies are likely to change their personal hygiene practices, food preparation behaviour and healthcare seeking practice following the hospitalisation of an affected member with an episode of diarrhoeal disease, which could also reduce their risk of suffering and/or hospitalisation for diarrhoea. While evidence of protective behavioural changes following a diarrhoea episode is scanty, prior literature has suggested that handwashing behaviour tends to increase during cholera and other outbreaks [42-44].

In this study, gender was not found to modify the temperature–diarrhoea relationship with males and females having nearly similar risk. However, the effect of mean temperature on diarrhoea was highest among children under 5 years of age. Compared to adults, young children may be more vulnerable to the effects of temperature owing to their immature immune system and low self-care capacity [45]. However, one global cross-sectional study did not find any significant relationship between temperature and diarrhoea in children under 5 years of age [46]. On the contrary, a study from Ethiopia reported significant positive effect of temperature on childhood diarrhoea [47]. Despite the potential vulnerability of children to climate change, empirical studies examining the role of age on the temperature–diarrhoea relationship are scarce. Future research examining the modulating impacts of age on climate–diarrhoea relationships can shed further light on this topic.

Our study has several strengths and limitations. The estimate of baseline impact of temperature on diarrhoea is based on high-quality, continuous data covering a 30-year period from the largest diarrhoea hospital in Dhaka, Bangladesh. In the context of a low- and middleincome country, the availability of such a dataset is rather exceptional. Most importantly, the analysis used temperature-diarrhoea relationship derived from a model that not only considered the effect of temperature on diarrhoea, but also the combined effects of temperature and other meteorological parameters including the immediate and lagged effects of humidity and rainfall. Furthermore, incorporation of WASH factors and other socio-economic indicators into the model during sensitivity analyses revealing comparable results further enhanced the confidence in our findings. While the use of sophisticated models to generate the temperaturediarrhoea relationships using long-term clinically diagnosed historical data along with an exhaustive sensitivity analysis, helped to narrow the uncertainty range associated with IRR estimates, some limitations remain. Hospitalisations for diarrhoeal diseases at the icddr,b Dhaka Hospital may not be representative of all affected diarrhoea cases from Dhaka. It is believed that for every patient seeking hospital care for diarrhoea, there are four patients seeking care at the community [48,49]. Hospital admissions, therefore, only represent a tip of the iceberg. Additionally, these represent the more severe cases. However, Dhaka Hospital is the largest diarrhoea treatment facility in the country and served a population of about one and a half million in 2010. Furthermore, icddr,b Dhaka Hospital services are likely to be accessed predominantly by the disadvantaged group and the surrounding slum dwellers. The estimates

from this study are therefore conservative and likely to underestimate the actual effect of temperature variations on diarrhoeal disease transmission in Dhaka. Nonetheless, severe cases from the less privileged group are important given that such cases have implications for healthcare utilisation and may require policy changes to allocate more resources to improve diarrhoea hospital care in Dhaka. Although less severe cases and the elite class are less likely to be included, the results of this study comparing temperature-diarrhoea relationship over time are still valid. Future studies using recent data may help to validate the findings from this study.

Furthermore, climate change impact calculations were based on global warming targets of 1.5–2°C by the 2100. However, as evident from several models from Bangladesh, this rise in temperature is likely to be reached much sooner and the temperature of Dhaka could increase by 4.3°C by the 2100. Although climate models are prone to biases, we calculated the likely impacts based on a few local climate projections and found that diarrhoea hospitalisations in all ages could increase by up to 15.9% for Dhaka. In addition, we assumed that the baseline relationship between meteorological factors and diarrhoea hospitalisation risk will remain constant in the future. However, future diarrhoea risk will be affected by many other factors including population density, access to healthcare and vaccines that we did not aim to model since we wanted to quantify climate change risks only. Although the role of access to safe water and sanitation were accounted for in the temperature-diarrhoea relationship estimates during the sensitivity analyses, the possible effects of other adaptive health policies on diarrhoea including increased access to vaccines targeting cholera and rotavirus on diarrhoea prevalence and hospitalisation could not be incorporated in these models due to lack of available data. However, seasonal adjustment accounted for the variations to some extent. While our models were adjusted for the immediate and lagged effects of heavy rainfall, we likely could not fully account for the potential effects of other factors (e.g. environmental, behavioural) linked specifically with the rainy monsoon season that may have also contributed to diarrhoea incidence in Dhaka. In the light of the above strengths and limitations, the predicted increase of 3.1–15.9% in diarrhoea hospitalisations with 0.91–4.3°C increase in temperature due to climate change should be considered indicative. Future studies using bias-corrected, temperature projections under additional climate change scenarios describing a range of changes in climate and related global warming, from mild to extreme, can generate better estimates of temperature-related excess diarrhoea for Dhaka.

Conclusions

This study demonstrated a statistically significant relationship between ambient temperature, relative humidity and heavy rainfall in the sub-tropical city of Dhaka. Given the high prevalence of diarrhoeal diseases in Bangladesh, even a small increase in risk stemming from increased environmental temperature in the context of climate change could lead to substantial burden and overwhelm the healthcare services in Dhaka. This underscores the importance of preparing the city for prevention and management of diarrhoeal diseases in anticipation of such increases in the future.

Supporting information

S1 File. Additional information on health data. (DOCX)

S2 File. Conceptual framework for confounders/covariates/effect modifiers. (DOCX)

S3 File. Sensitivity analyses and model diagnostics. (DOCX)

S4 File. Definitions/descriptions of climate terms and jargon. (DOCX)

S1 Data. Excel spreadsheet containing, in separate sheets, the underlying numerical data and statistical analysis for Figs 1–5. (XLSX)

Acknowledgments

The authors would like to thank members of the Research Administration team of icddr,b, National Health Foundation and Research Institute (NHF&RI) and Bangladesh Meteorological Department (BMD) for providing data for the study.

Author Contributions

Conceptualization: Farhana Haque.

Data curation: Farhana Haque.

Formal analysis: Farhana Haque.

Funding acquisition: Farhana Haque.

Investigation: Farhana Haque.

Methodology: Farhana Haque, Shakoor Hajat, Ilan Kelman.

Project administration: S. M. Tafsir Hasan, A. S. G. Faruque, Tahmeed Ahmed, Shamim Jubayer.

Resources: Farhana Haque, Ilan Kelman.

Supervision: Fiona C. Lampe, Shakoor Hajat, Katerina Stavrianaki, Ilan Kelman.

Validation: Farhana Haque.

Writing - original draft: Farhana Haque.

Writing – review & editing: Farhana Haque, Fiona C. Lampe, Shakoor Hajat, Katerina Stavrianaki, S. M. Tafsir Hasan, A. S. G. Faruque, Tahmeed Ahmed, Shamim Jubayer, Ilan Kelman.

References

- IPCC. Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, 2021.
- Cook J, Oreskes N, Doran PT, Anderegg WRL, Verheggen B, Maibach E, et al. Consensus on consensus: a synthesis of consensus estimates on human-caused global warming. Environmental Research Letter. 2016; 11 (4). https://doi.org/10.1088/1748-9326/11/4/048002
- Watts N, Amann M, Arnell N, Ayeb-Karlsson S, Belesova K, Berry H, et al. The 2018 report of the Lancet Countdown on health and climate change: shaping the health of nations for centuries to come. Lancet. 2018; 392(10163):2479–514. Epub 2018/12/07. https://doi.org/10.1016/S0140-6736(18)32594-7 PMID: 30503045.
- Whitmee S, Haines A, Beyrer C, Boltz F, Capon AG, de Souza Dias BF, et al. Safeguarding human health in the Anthropocene epoch: report of The Rockefeller Foundation-Lancet Commission on planetary health. Lancet. 2015; 386(10007):1973–2028. Epub 2015/07/21. https://doi.org/10.1016/S0140-6736(15)60901-1 PMID: 26188744.

- Levy K, Smith SM, Carlton EJ. Climate Change Impacts on Waterborne Diseases: Moving Toward Designing Interventions. Curr Environ Health Rep. 2018; 5(2):272–82. Epub 2018/05/04. https://doi. org/10.1007/s40572-018-0199-7 PMID: 29721700; PubMed Central PMCID: PMC6119235.
- Mellor JE, Levy K, Zimmerman J, Elliott M, Bartram J, Carlton E, et al. Planning for climate change: The need for mechanistic systems-based approaches to study climate change impacts on diarrheal diseases. Sci Total Environ. 2016;548–549:82–90. Epub 2016/01/23. https://doi.org/10.1016/j.scitotenv. 2015.12.087 PMID: 26799810; PubMed Central PMCID: PMC4818006.
- Collaborators GBDDD. Estimates of global, regional, and national morbidity, mortality, and aetiologies of diarrhoeal diseases: a systematic analysis for the Global Burden of Disease Study 2015. Lancet Infect Dis. 2017; 17(9):909–48. Epub 2017/06/06. https://doi.org/10.1016/S1473-3099(17)30276-1 PMID: 28579426; PubMed Central PMCID: PMC5589208.
- 8. Campbell-Lendrum DH, Corvalán CF, Prüss–Ustün A, editors. How much disease could climate change cause? Geneva: World Health Organization; 2003.
- Kolstad EW, Johansson KA. Uncertainties associated with quantifying climate change impacts on human health: a case study for diarrhea. Environ Health Perspect. 2011; 119(3):299–305. Epub 2010/ 10/12. https://doi.org/10.1289/ehp.1002060 PMID: 20929684; PubMed Central PMCID: PMC3059990.
- Moors E, Singh T, Siderius C, Balakrishnan S, Mishra A. Climate change and waterborne diarrhoea in northern India: impacts and adaptation strategies. Sci Total Environ. 2013;468–469 Suppl:S139-51. Epub 2013/08/27. https://doi.org/10.1016/j.scitotenv.2013.07.021 PMID: 23972324.
- Singh RB, Hales S, de Wet N, Raj R, Hearnden M, Weinstein P. The influence of climate variation and change on diarrheal disease in the Pacific Islands. Environ Health Perspect. 2001; 109(2):155–9. Epub 2001/03/27. https://doi.org/10.1289/ehp.01109155 PMID: 11266326; PubMed Central PMCID: PMC1240636.
- Checkley W, Epstein LD, Gilman RH, Figueroa D, Cama RI, Patz JA, et al. Effect of El Nino and ambient temperature on hospital admissions for diarrhoeal diseases in Peruvian children. Lancet. 2000; 355 (9202):442–50. Epub 2000/06/07. https://doi.org/10.1016/s0140-6736(00)82010-3 PMID: 10841124.
- Hajat S, Gampe D, Sarsour A, Abuzerr S. Climate Change and Diarrhoeal Disease Burdens in the Gaza Strip, Palestine: Health Impacts of 1.5 degrees C and 2 degrees C Global Warming Scenarios. Int J Environ Res Public Health. 2022; 19(8). Epub 20220418. https://doi.org/10.3390/ijerph19084898 PMID: 35457766; PubMed Central PMCID: PMC9026927.
- 14. Islam HMT, Kamruzzaman M, Shahid S, Mainuddin M, Alam E, Islam ARMT, et al. Spatiotemporal changes in temperature projections over Bangladesh using multi-model ensemble data. Frontiers in Environmental Science. 2023;10. https://doi.org/10.3389/fenvs.2022.1074974.
- Anik AH, Sultan MB, Alam M, Parvin F, Ali MM, Tareq SM. The impact of climate change on water resources and associated health risks in Bangladesh: A review. Water Security. 2023;18. <u>https://doi.org/10.1016/j.wasec.2023.100133</u>
- Islam MS, Okubo K, Islam AHMS, Sato M. Investigating the effect of climate change on food loss and food security in Bangladesh. SN Bus Econ. 2022; 2(4). https://doi.org/10.1007/s43546-021-00177-z.
- Charles KJ, Howard G, Prats EV, Gruber J, Alam S, Alamgir ASM, et al. Infrastructure alone cannot ensure resilience to weather events in drinking water supplies. Sci Total Environ. 2021:151876. Epub 2021/11/27. https://doi.org/10.1016/j.scitotenv.2021.151876 PMID: 34826465.
- Haque F, Hossain MJ, Kundu SK, Naser AM, Rahman M, Luby SP. Cholera Outbreaks in Urban Bangladesh In 2011. Epidemiology (Sunnyvale). 2013;3. Epub 2013/01/01. https://doi.org/10.4172/2161-1165.1000126 PMID: 26702366; PubMed Central PMCID: PMC4686147.
- UN-Habitat. State of the World's Cities 2008/2009—Harmonious Cities. London, UK and Sterling, VA, USA: United Nations Human Settlements Programme, 2008 ISBN: 978-1-84407-696-3.
- Mowla Q, Islam M. Natural Drainage System and Water Logging in Dhaka: Measures to address the Problems. J Bangladesh Inst Plan. 2013; 6:23–33.
- Hassan MM, Southworth J. Analyzing Land Cover Change and Urban Growth Trajectories of the Mega-Urban Region of Dhaka Using Remotely Sensed Data and an Ensemble Classifier. Sustainability. 2018; 10(10). https://doi.org/10.3390/su10010010
- 22. Swapan MSH, Zaman AU, Ahsan T, Ahmed F. Transforming Urban Dichotomies and Challenges of South Asian Megacities: Rethinking Sustainable Growth of Dhaka, Bangladesh. Urban Sci. 2017;1(31). https://doi.org/10.3390/urbansci1040031
- 23. UN. World Population Prospects: The 2017 Revision, Key Findings and Advance Tables. New York: Department of Economic and Social Affairs, Population Division, 2017.
- Stoll BJ, Glass RI, Huq MI, Khan MU, Holt JE, Banu H. Surveillance of patients attending a diarrhoeal disease hospital in Bangladesh. Br Med J (Clin Res Ed). 1982; 285(6349):1185–8. Epub 1982/10/23. https://doi.org/10.1136/bmj.285.6349.1185 PMID: 6812801; PubMed Central PMCID: PMC1500105.

- Haque F, Lampe F, Hajat S, Stavrianaki K, Hasan SMT, Faruque A, et al. Effects of diurnal temperature range on diarrhea in the subtropical megacity of Dhaka, Bangladesh. The Journal of Climate Change and Health. 2024;17. https://doi.org/10.1016/j.joclim.2024.100305
- Alamgir M, Ahmed K, Homsi R, Dewan A, Wang J-J, Shahid S. Downscaling and Projection of Spatiotemporal Changes in Temperature of Bangladesh. Earth Systems and Environment 2019;3. https://doi.org/10.1007/s41748-019-00121-0
- 27. The Cost of Adapting to Extreme Weather Events in a Changing Climate. Dhaka The World Bank 2011.
- Rahman MM, Islam MN, Ahmed AU, Georgi F. Rainfall and temperature scenarios for Bangladesh for the middle of 21st century using RegCM. J Earth Syst Sci 2012; 121:287–95. https://doi.org/10.1007/ s12040-012-0159-9
- 29. Climate: Observations, projections and impacts. Exeter, Devon, United Kingdom: UK Met Office Hadley Centre 2011.
- **30.** Shourav MSA, Mohsenipour M, Alamgir M, Pour SH, Ismail T. Historical trends and future projections of climate at Dhaka City of Bangladesh Jurnal Teknologi (Sciences & Engineering). 2016; 78:69–75.
- Bhaskaran K, Gasparrini A, Hajat S, Smeeth L, Armstrong B. Time series regression studies in environmental epidemiology. Int J Epidemiol. 2013; 42(4):1187–95. Epub 2013/06/14. https://doi.org/10.1093/ije/dyt092 PMID: 23760528; PubMed Central PMCID: PMC3780998.
- **32.** Hardin JW, Hilbe JM, editors. Generalized Linear Models and Extensions 2nd ed. Texas 77845: Stata Press; 2007.
- Carlton EJ, Woster AP, DeWitt P, Goldstein RS, Levy K. A systematic review and meta-analysis of ambient temperature and diarrhoeal diseases. Int J Epidemiol. 2016; 45(1):117–30. Epub 2015/11/15. https://doi.org/10.1093/ije/dyv296 PMID: 26567313; PubMed Central PMCID: PMC4881833.
- 34. Billah SM, Raihana S, Ali NB, Iqbal A, Rahman MM, Khan ANS, et al. Bangladesh: a success case in combating childhood diarrhoea. J Glob Health. 2019; 9(2):020803. Epub 2019/11/02. https://doi.org/10.7189/jogh.09.020803 PMID: 31673347; PubMed Central PMCID: PMC6816141 www.icmje.org/coi_disclosure.pdf (available upon request from the corresponding author), and declare no conflicts of interest.
- 35. Black R, Fontaine O, Lamberti L, Bhan M, Huicho L, El Arifeen S, et al. Drivers of the reduction in child-hood diarrhea mortality 1980–2015 and interventions to eliminate preventable diarrhea deaths by 2030. J Glob Health. 2019; 9(2):020801. Epub 2019/11/02. https://doi.org/10.7189/jogh.09.020801 PMID: 31673345; PubMed Central PMCID: PMC6815873 provided funding for the analyses. The authors completed the Unified Competing Interest form at www.icmje.org/coi_disclosure.pdf (available upon request from the corresponding author), and declare no further conflicts of interest.
- Haque F. Impacts of temperature, diurnal temperature range, heat index and heat wave on diarrhoeal diseases in Dhaka, Bangladesh in the context of climate change: A time-series analysis [PhD Thesis]. London University College London (UCL); 2022.
- Levy K, Woster AP, Goldstein RS, Carlton EJ. Untangling the Impacts of Climate Change on Waterborne Diseases: a Systematic Review of Relationships between Diarrheal Diseases and Temperature, Rainfall, Flooding, and Drought. Environ Sci Technol. 2016; 50(10):4905–22. Epub 2016/04/09. https:// doi.org/10.1021/acs.est.5b06186 PMID: 27058059; PubMed Central PMCID: PMC5468171.
- Azman AS, Rudolph KE, Cummings DA, Lessler J. The incubation period of cholera: a systematic review. J Infect. 2013; 66(5):432–8. Epub 20121129. <u>https://doi.org/10.1016/j.jinf.2012.11.013</u> PMID: 23201968; PubMed Central PMCID: PMC3677557.
- Andrews JR, Leung DT, Ahmed S, Malek MA, Ahmed D, Begum YA, et al. Determinants of severe dehydration from diarrheal disease at hospital presentation: Evidence from 22 years of admissions in Bangladesh. PLoS Negl Trop Dis. 2017; 11(4):e0005512. Epub 2017/04/28. https://doi.org/10.1371/ journal.pntd.0005512 PMID: 28448489; PubMed Central PMCID: PMC5423662.
- Aik J, Ong J, Ng LC. The effects of climate variability and seasonal influence on diarrhoeal disease in the tropical city-state of Singapore—A time-series analysis. Int J Hyg Environ Health. 2020; 227:113517. Epub 2020/04/10. https://doi.org/10.1016/j.ijheh.2020.113517 PMID: 32272437.
- Phung D, Chu C, Rutherford S, Nguyen HLT, Luong MA, Do CM, et al. Heavy rainfall and risk of infectious intestinal diseases in the most populous city in Vietnam. Sci Total Environ. 2017; 580:805–12. Epub 2016/12/26. https://doi.org/10.1016/j.scitotenv.2016.12.027 PMID: 28012659.
- 42. White S, Mutula AC, Buroko MM, Heath T, Mazimwe FK, Blanchet K, et al. How does handwashing behaviour change in response to a cholera outbreak? A qualitative case study in the Democratic Republic of the Congo. PLoS One. 2022; 17(4):e0266849. Epub 20220412. https://doi.org/10.1371/journal.pone.0266849 PMID: 35413080; PubMed Central PMCID: PMC9004767.
- 43. White S, Thorseth AH, Dreibelbis R, Curtis V. The determinants of handwashing behaviour in domestic settings: An integrative systematic review. Int J Hyg Environ Health. 2020; 227:113512. Epub 20200325. https://doi.org/10.1016/j.ijheh.2020.113512 PMID: 32220763.

- 44. Zohura F, Bhuyian SI, Monira S, Begum F, Biswas SK, Parvin T, et al. Observed Handwashing with Soap Practices Among Cholera Patients and Accompanying Household Members in a Hospital Setting (CHoBI7 Trial). Am J Trop Med Hyg. 2016; 95(6):1314–8. Epub 20161003. https://doi.org/10.4269/ ajtmh.16-0379 PMID: 27698272; PubMed Central PMCID: PMC5154445.
- 45. Xu Z, Huang C, Turner LR, Su H, Qiao Z, Tong S. Is diurnal temperature range a risk factor for childhood diarrhea? PLoS One. 2013; 8(5):e64713. Epub 2013/06/01. https://doi.org/10.1371/journal.pone. 0064713 PMID: 23724083; PubMed Central PMCID: PMC3665771.
- Lloyd SJ, Kovats RS, Armstrong BG. Global diarrhoea morbidity, weather and climate. Clim Res. 2007; 34: 119–27.
- Azage M, Kumie A, Worku A, A CB, Anagnostou E. Effect of climatic variability on childhood diarrhea and its high risk periods in northwestern parts of Ethiopia. PLoS One. 2017; 12(10):e0186933. Epub 2017/10/27. https://doi.org/10.1371/journal.pone.0186933 PMID: 29073259; PubMed Central PMCID: PMC5658103.
- Nhampossa T, Mandomando I, Acacio S, Nhalungo D, Sacoor C, Nhacolo A, et al. Health care utilization and attitudes survey in cases of moderate-to-severe diarrhea among children ages 0–59 months in the District of Manhica, southern Mozambique. Am J Trop Med Hyg. 2013; 89(1 Suppl):41–8. Epub 2013/05/01. https://doi.org/10.4269/ajtmh.12-0754 PMID: 23629927; PubMed Central PMCID: PMC3748500.
- 49. Sultana R, Luby SP, Gurley ES, Rimi NA, Swarna ST, Khan JAM, et al. Cost of illness for severe and non-severe diarrhea borne by households in a low-income urban community of Bangladesh: A crosssectional study. PLoS Negl Trop Dis. 2021; 15(6):e0009439. Epub 2021/06/12. https://doi.org/10.1371/ journal.pntd.0009439 PMID: 34115764; PubMed Central PMCID: PMC8221788.